Check for updates

Citation: Bann D, Johnson W, Li L, Kuh D, Hardy R (2017) Socioeconomic Inequalities in Body Mass Index across Adulthood: Coordinated Analyses of Individual Participant Data from Three British Birth Cohort Studies Initiated in 1946, 1958 and 1970. PLoS Med 14(1): e1002214. doi:10.1371/journal. pmed.1002214

Academic Editor: Sanjay Basu, Stanford University, UNITED STATES

Received: September 1, 2016

Accepted: December 1, 2016

Published: January 10, 2017

Copyright: © 2017 Bann et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The dataset used in this research uses harmonised data from three cohort studies. The original and the harmonised 1946 NSHD data are made available to researchers who submit data requests to mrclha.swiftinfo@ucl. ac.uk; see also the full policy documents at http:// www.nshd.mrc.ac.uk/data.aspx. The original data for the 1958 NCDS and 1970 BCS are available from the UK Data Archive (http://www.dataarchive.ac.uk). **RESEARCH ARTICLE**

Socioeconomic Inequalities in Body Mass Index across Adulthood: Coordinated Analyses of Individual Participant Data from Three British Birth Cohort Studies Initiated in 1946, 1958 and 1970

David Bann¹*, William Johnson², Leah Li³, Diana Kuh⁴, Rebecca Hardy⁴

1 Centre for Longitudinal Studies, UCL Institute of Education, London, United Kingdom, 2 School of Sport, Exercise and Health Sciences, Loughborough University, Loughborough, United Kingdom, 3 Population, Policy and Practice, UCL Institute of Child Health, London, United Kingdom, 4 MRC Unit for Lifelong Health and Ageing at UCL, London, United Kingdom

* david.bann@ucl.ac.uk

Abstract

Background

High body mass index (BMI) is an important contributor to the global burden of ill-health and health inequality. Lower socioeconomic position (SEP) in both childhood and adulthood is associated with higher adult BMI, but how these associations have changed across time is poorly understood. We used longitudinal data to examine how childhood and adult SEP relates to BMI across adulthood in three national British birth cohorts.

Methods and Findings

The sample comprised up to 22,810 participants with 77,115 BMI observations in the 1946 MRC National Survey of Health and Development (ages 20 to 60–64), the 1958 National Child Development Study (ages 23 to 50), and the 1970 British Cohort Study (ages 26 to 42). Harmonized social class-based SEP data (Registrar General's Social Class) was ascertained in childhood (father's class at 10/11 y) and adulthood (42/43 years), and BMI repeatedly across adulthood, spanning 1966 to 2012. Associations between SEP and BMI were examined using linear regression and multilevel models.

Lower childhood SEP was associated with higher adult BMI in both genders, and differences were typically larger at older ages and similar in magnitude in each cohort. The strength of association between adult SEP and BMI did not vary with age in any consistent pattern in these cohorts, but were more evident in women than men, and inequalities were larger among women in the 1970 cohort compared with earlier-born cohorts. For example, mean differences in BMI at 42/43 y amongst women in the lowest compared with highest social class were 2.0 kg/m² (95% CI: -0.1, 4.0) in the 1946 NSHD, 2.3 kg/m² (1.1, 3.4) in the 1958 NCDS, and 3.9 kg/m² (2.3, 5.4) the in the 1970 BCS; mean (SD) BMI in the highest and lowest social classes were as follows: 24.9 (0.8) versus 26.8 (0.7) in the 1946 NSHD,

Funding: This project is part of a collaborative research programme entitled 'Cohorts and Longitudinal Studies Enhancement Resources' (CLOSER). This programme is funded by the ESRC (http://www.esrc.ac.uk) grant reference: ES/ K000357/1. The UK Medical Research Council (http://www.mrc.ac.uk) provides core funding for the MRC National Survey of Health and Development and RH and DK (MC_UU_12019/1, MC_UU_12019/2). The UK Economic and Social Research Council provides core funding for the National Child Development Study and the British Cohort Study (ES/M001660/1). DB is partly funded by the ESRC (grant reference ES/M008584/1), but has no involvement in funding decisions from ESRC. RH and DK work at the MRC Unit for Lifelong Health and Ageing at UCL but have no involvement in funding decisions for the MRC. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Abbreviations: BMI, body mass index; SD, standard deviation; SE, standard error; SEP, socioeconomic position. 24.2 (0.4) versus 26.5 (0.4) in the 1958 NCDS, and 24.2 (0.3) versus 28.1 (0.8) in the 1970 BCS. Findings did not differ whether using overweight or obesity as an outcome.

Limitations of this work include the use of social class as the sole indicator of SEP—while it was available in each cohort in both childhood and adulthood, trends in BMI inequalities may differ according to other dimensions of SEP such as education or income. Although harmonized data were used to aid inferences about birth cohort differences in BMI inequality, differences in other factors may have also contributed to findings—for example, differences in missing data.

Conclusions

Given these persisting inequalities and their public health implications, new and effective policies to reduce inequalities in adult BMI that tackle inequality with respect to both childhood and adult SEP are urgently required

Author Summary

Why Was This Study Done?

- High body mass index (BMI) is thought to be harmful to human health—in most adults, a high BMI is due to having high amounts of fat mass in the body.
- Previous studies have found that those with fewer socioeconomic resources—both as children and as adults—are more likely, on average, to have a higher BMI as adults.
- Reducing these socioeconomic inequalities in BMI is an important health policy goal, yet there is limited existing data to help us understand comprehensively how these inequalities have changed across time.

What Did the Researchers Do and Find?

- We used data from three national British birth cohort studies of those born in 1946, 1958, and 1970—these studies contain comparable data on social class in childhood and adulthood, and on BMI across adult life.
- We confirm that large inequalities in BMI exist, according to both childhood and adult SEP—these were stronger among women, but also found among men.
- Inequalities according to childhood SEP generally become progressively larger at older ages in all cohorts and in both genders; inequalities according to adult SEP were larger among more recently born generations of women.

What Do These Findings Mean?

• The fact that BMI inequalities have persisted or increased across different generations, despite policies designed to reduce them, suggests that new policies are required.

- Results support the need to intervene earlier rather than later in adult life, since inequalities tend to become larger at older ages.
- Limitations include the use of only one aspect of socioeconomic circumstances (social class), and the fact that not all participants continue to provide data in longitudinal studies—this may have led us to underestimating the size of BMI inequalities.

Introduction

High body mass index (BMI) is an important modifiable contributor to the global burden of ill-health and health expenditure, [1, 2] and its prevalence increased markedly between the 1980s and 2014.[3–5] National attempts to reduce population BMI levels have thus far largely been unsuccessful, [3–5] suggesting that it is likely to be an important threat to the health of future generations.[3] Indeed, the increasing prevalence of high BMI at younger ages suggests that later-born generations are at risk of spending longer periods of life either overweight or obese.[6] Systematic reviews have also shown that in high income countries, lower socioeconomic position (SEP) in childhood and adulthood are associated with higher adult BMI and increased obesity risk.[7–9] Due to the links between higher BMI and adverse health outcomes, [10, 11] inequalities in BMI are likely to be an important contributor to socioeconomic inequalities in health. Accordingly, reducing BMI inequalities is a stated goal of numerous health policymakers and organizations; [12] achieving this requires high-quality evidence on how such inequalities have changed across time, in response to changing policy and economic contexts.

Existing evidence on how inequalities in BMI have changed across generations is largely restricted to repeated analyses of cross-sectional data.[13-21] These studies have tended to report persisting inequalities in BMI that are stronger among women, yet are limited to investigating relatively recent changes (e.g., from 1993/1994 to 2002/2003,[13] or 1994 to 2008[14]). They also do not elucidate how inequalities in BMI change with age; understanding these patterns may help inform the development of interventions targeted at the most effective ages. Such differences may reflect age differences in susceptibility to weight gain, which in turn may differ by cohort depending on the period of life when exposed to more obesogenic environments. The use of repeated cross-sectional adult data also leads to an almost exclusive focus on the changing consequences of adult SEP in previous studies, since such studies do not have (by design) prospective measures of childhood SEP. However, childhood socioeconomic circumstances strongly determine those in adulthood, and childhood SEP has been repeatedly related to higher adult adiposity and other health outcomes independent of adult SEP.[22-27] Relations between childhood SEP and adult BMI are also less likely to be affected by reverse causation than those with adult SEP, since obesity may impair adult economic outcomes; [28–31] childhood SEP is therefore an important dimension of socioeconomic circumstances to consider when understanding how inequalities have changed across time. Many previous studies have not investigated inequality on both relative (e.g., % change) and absolute (difference in kg/m²) scales. Since changes in relative inequality can occur despite opposing or no changes in absolute inequality, particularly when dichotomized outcomes are used and the overall population outcome prevalence changes, analyzing both is likely to be important in order to better understand how inequalities have changed across time, and their population health impact. [32, 33] To our knowledge, the literature currently lacks a comprehensive coordinated analysis

to investigate how inequalities in BMI across adulthood have changed over the course of the obesity epidemic, with respect to both childhood and adult SEP.

The objectives of this study were to examine trends in the socioeconomic distribution of BMI and overweight or obesity across adult life, using data from the British birth cohort studies initiated in 1946, 1958, and 1970. These data have previously been used to show that the obesity epidemic has hit more recently born generations at increasingly younger ages in adult-hood;[6] as with evidence from repeated cross-sectional studies,[4] this paper found that obesity prevalence increased markedly from the 1980s onwards in the United Kingdom and remained persistently high up to 2012. We use data collected between 1966 and 2012, which covers this period. Consistent with a fundamental cause hypothesis for understanding health inequality,[34, 35] we expected that due to the increasing public dissemination of the health harms of obesity, those of higher SEP may have been increasingly able to use their greater social, financial, and educational resources to protect themselves against excessive weight gain across adulthood; principally by modifying their diet and/or physical activity levels. As such, we hypothesized that socioeconomic inequalities in BMI would be larger in cohorts born later in the 20th century, and that these differences would be evident for both childhood and adult SEP.

Methods

Study samples

Each study used in this manuscript has received relevant ethical approval and obtained parental/participant consent; this information is available from the study websites and/or cohort profiles.

Britain's birth cohort studies with participants followed up through adulthood were used. These were designed to be nationally representative when initiated in 1946 (MRC National Survey of Health and Development [36, 37]-1946 NSHD), 1958 (National Child Development Study[38]-1958 NCDS), and 1970 (British Cohort Study[39]-1970 BCS). The history, design, and characteristics of these studies have been previously described in detail in papers [36-40] and books; [41, 42] studies have also examined the characteristics of those lost due to attrition.[43-46] To aid the comparability of associations between SEP and BMI across studies, analyses were restricted to singleton births in England, Scotland, and Wales from those born and included in cohorts in the relevant weeks in March/April 1946 (N = 5,362), 1958 (N = 16,383), and 1970 (N = 16,172). In the 1946 NSHD, participants were restricted to those from married mothers due to tracing difficulties in the minority of babies born to unmarried mothers.[40] The weeks of initiation were chosen on the basis of practical considerations at the time of study development. The analytic sample sizes were those with valid data for each SEP measure and at least one adult BMI measure; for childhood SEP, a total of 22,810 participants with 77,115 BMI observations; for adult SEP, a total of 17,898 participants with 36,702 BMI observations (sample sizes for all analyses, in each cohort, age, and gender are shown in Table 1 and Table 2, and in S1 Table and S2 Table). All participants in the 1946 NSHD were white, as were 98.7% of those in the 1958 NCDS, and 95.4% of 1970 BCS. All analyses using the 1946 NSHD were weighted to account for the stratified sampling design[37].

BMI measurement

BMI (kg/m²), the main outcome measure, was derived in each study from measured or self-reported weight and height obtained at all available adulthood ages: 1946 NSHD: 20*, 26*, 36, 43, 53, and 60–64 y; 1958 NCDS: 23*, 33, 42*, 44, and 50* y; 1970 BCS: 26*, 30*, 34*, and 42* y (*self-report). The main protocol differences and methods used to harmonize height and

	Gender, age		kg/m ² BMI difference (95% CI)					
Cohort								
1946 NSHD	Men	N	I (ref)	II	III NM	III M	IV	V
	20	1,562	-	0.6 (0.0, 1.2)	0.3 (-0.3, 0.9)	0.9 (0.3, 1.4)	0.9 (0.3, 1.5)	1.4 (0.7, 2.2)
	26	1,552	-	0.7 (0.1, 1.3)	0.6 (-0.1, 1.3)	1.2 (0.6, 1.7)	1.3 (0.6, 2.0)	1.7 (0.9, 2.5)
	36	1,365	-	0.9 (0.1, 1.8)	0.7 (-0.2, 1.6)	1.6 (0.8, 2.4)	1.8 (0.9, 2.7)	1.5 (0.4, 2.6)
	43	1,338	- #	0.9 (0.1, 1.8)	0.8 (-0.1, 1.7)	1.7 (0.9, 2.4)	1.9 (1.0, 2.8)	1.3 (0.2, 2.3)
	53	1,206	- #	1.0 (0.0, 2.0)	0.7 (-0.3, 1.8)	1.9 (1.0, 2.9)	2.2 (1.1, 3.3)	1.5 (0.3, 2.8)
	60–64	898	- #	0.8 (-0.4, 2.0)	0.7 (-0.5, 1.8)	2.2 (1.2, 3.3)	2.3 (1.0, 3.5)	1.1 (-0.5, 2.8)
1958 NCDS	23	4,046	-	0.4 (-0.1, 0.8)	0.3 (-0.2, 0.8)	0.8 (0.4, 1.3)	0.8 (0.3, 1.3)	0.9 (0.4, 1.4)
	33	3,547	-	0.3 (-0.3, 1.0)	0.6 (0.0, 1.3)	0.9 (0.3, 1.5)	1.0 (0.4, 1.7)	0.9 (0.2, 1.6)
	42	3,637	-	0.7 (0.0, 1.3)	0.7 (0.0, 1.4)	1.0 (0.4, 1.6)	1.2 (0.5, 1.9)	1.4 (0.7, 2.2)
	44	3,046	-	0.7 (-0.1, 1.4)	0.8 (0.0, 1.7)	1.2 (0.5, 1.9)	1.6 (0.8, 2.4)	1.8 (0.9, 2.6)
	50	2,766	-	1.0 (0.1, 1.8)	1.1 (0.2, 2.1)	1.6 (0.8, 2.4)	2.2 (1.3, 3.1)	2.1 (1.2, 3.1)
1970 BCS	26	1,923	-	0.3 (-0.4, 1.0)	0.0 (-0.8, 0.7)	0.8 (0.1, 1.4)	1.1 (0.4, 1.9)	1.2 (0.3, 2.1)
	30	3,960	-	0.4 (-0.1, 1.0)	0.3 (-0.4, 0.9)	0.8 (0.3, 1.3)	0.6 (0.0, 1.2)	0.7 (0.0, 1.4)
	34	3,435	-	0.4 (-0.2, 1.0)	0.4 (-0.3, 1.1)	1.0 (0.4, 1.6)	1.1 (0.4, 1.8)	1.2 (0.4, 1.9)
	42	3,208	-	0.4 (-0.4, 1.1)	0.6 (-0.2, 1.5)	1.3 (0.6, 2.0)	1.3 (0.5, 2.1)	0.9 (0.0, 1.9)
1946 NSHD	Women	N	I (ref)	II	III NM	III M	IV	V
	20	1,426	-	0.3 (-0.4, 1.0)	-0.2 (-0.8, 0.5)	0.3 (-0.3, 0.9)	0.8 (0.1, 1.4)	0.9 (0.0, 1.9)
	26	1,543	-	0.5 (-0.2, 1.2)	0.3 (-0.4, 0.9)	0.9 (0.3, 1.5)	1.4 (0.7, 2.1)	1.7 (0.6, 2.7)
	36	1,353	- #	0.5 (-0.4, 1.4)	-0.1 (-0.9, 0.8)	1.3 (0.5, 2.2)	2.0 (1.0, 2.9)	1.7 (0.5, 2.9)
	43	1,315	-	0.5 (-0.6, 1.6)	0.2 (-0.8, 1.2)	1.4 (0.4, 2.4)	2.2 (1.1, 3.3)	1.7 (0.2, 3.2)
	53	1,235	-	0.8 (-0.5, 2.2)	0.5 (-0.9, 1.8)	1.7 (0.4, 2.9)	3.0 (1.6, 4.4)	1.6 (-0.4, 3.6)
	60–64	958	- #	1.1 (-0.3, 2.6)	0.2 (-1.2, 1.6)	1.6 (0.3, 2.8)	3.6 (2.1, 5.1)	3.1 (0.6, 5.7)
1958 NCDS	23	4,115	- #	0.6 (0.1, 1.2)	0.5 (0.0, 1.1)	1.4 (0.9, 1.9)	1.3 (0.8, 1.9)	1.2 (0.6, 1.8)
	33	3,524	- #	0.7 (-0.2, 1.5)	0.9 (0.0, 1.8)	1.9 (1.1, 2.7)	1.9 (1.0, 2.8)	1.3 (0.3, 2.3)
	42	3,668	- #	0.8 (-0.1, 1.6)	0.8 (-0.1, 1.7)	2.0 (1.2, 2.8)	2.3 (1.4, 3.1)	1.5 (0.6, 2.5)
	44	3099	- #	0.6 (-0.4, 1.6)	0.9 (-0.2, 2.0)	1.9 (1.0, 2.9)	2.1 (1.1, 3.2)	1.1 (0.0, 2.3)
	50	2,759	- #	0.4 (-0.6, 1.5)	1.0 (-0.1, 2.1)	2.0 (1.0, 3.0)	2.2 (1.1, 3.3)	0.7 (-0.5, 1.9)
1970 BCS	26	3,545	-	0.6 (0.0, 1.2)	0.4 (-0.3, 1.1)	1.2 (0.6, 1.8)	1.3 (0.6, 1.9)	1.5 (0.7, 2.2)
	30	4,132	-	0.7 (0.0, 1.4)	0.7 (0.0, 1.5)	1.6 (1.0, 2.3)	1.8 (1.0, 2.5)	2.2 (1.4, 3.1)
	34	3,642	-	0.3 (-0.4, 1.1)	0.4 (-0.4, 1.3)	1.3 (0.6, 2.1)	1.4 (0.6, 2.3)	2.7 (1.7, 3.7)
	42	3,312	- #	0.3 (-0.6, 1.2)	0.1 (-0.9, 1.2)	1.7 (0.9, 2.6)	1.6 (0.6, 2.6)	2.7 (1.6, 3.9)

Table 1. Father's social class (10/11 y) and BMI across adulthood in the 1946 NSHD, the 1958 NCDS, and the 1970 BCS British birth cohort studies

#evidence for deviation from linearity—*p* < 0.05 for Wald tests; BMI differences estimated using linear regression models; occupational-based social classes were assigned as follows: I (professional), II (managerial and technical), III NM (skilled nonmanual), III M (skilled manual), IV (partly-skilled) V (unskilled)

doi:10.1371/journal.pmed.1002214.t001

weight have been described elsewhere.[6] Briefly, all measures were converted to metric units, women were excluded when pregnant (N = 257 (1946), 684 (1958), 110 (1970)), and a standardized cleaning process was used to remove participants with implausible values.

SEP ascertainment

Indicators of SEP in childhood and adulthood were derived in each cohort—the main exposures in this study. Social class measures were used, since comparable measures were available in both childhood and adulthood, and in each cohort. Childhood SEP was indicated by father's occupational social class, measured at 10/11 y, and adult SEP by own occupational social class

	Gender, age		kg/m ² E	MI difference (95%	CI)			
Cohort	Men	N	I (ref)	11	III NM	III M	IV	V
1946 NSHD	43	1,520	- #	0.8 (0.1, 1.4)	1.0 (0.1, 1.9)	1.1 (0.4, 1.8)	0.5 (-0.4, 1.4)	0.0 (-1.5, 1.5)
	53	1,283	-	0.3 (-0.5, 1.1)	0.3 (-0.9, 1.5)	0.8 (-0.1, 1.6)	0.0 (-1.1, 1.2)	-0.9 (-2.5, 0.7)
	60–64	963	-	0.2 (-0.8, 1.1)	0.7 (-0.6, 2.0)	1.3 (0.3, 2.3)	0.4 (-1.0, 1.8)	0.5 (-3.6, 4.6)
1958 NCDS	42	4,623	-	0.7 (0.3, 1.2)	0.9 (0.4, 1.4)	0.9 (0.4, 1.3)	1.0 (0.4, 1.5)	0.8 (0.0, 1.5)
	44	3,808	- #	0.7 (0.2, 1.2)	0.8 (0.1, 1.4)	1.0 (0.5, 1.5)	1.3 (0.6, 2.0)	0.3 (-0.7, 1.3)
	50	3,283	-	0.9 (0.4, 1.5)	1.2 (0.5, 1.9)	1.6 (1.0, 2.2)	1.9 (1.2, 2.7)	1.2 (0.0, 2.4)
1970 BCS	42	3,636	-#	0.8 (0.2, 1.4)	1.0 (0.3, 1.8)	1.4 (0.8, 2.0)	1.0 (0.3, 1.7)	0.4 (-0.7, 1.5)
	Women							
1946 NSHD	43	1,419	-	-0.5 (-2.0, 1.1)	0.1 (-1.5, 1.6)	0.6 (-1.3, 2.6)	1.6 (-0.1, 3.4)	2.0 (-0.1, 4.0)
	53	1,264	-	0.0 (-2.4, 2.4)	0.4 (-1.9, 2.8)	0.5 (-2.2, 3.2)	1.9 (-0.6, 4.4)	2.3 (-0.5, 5.1)
	60–64	999	-	-0.1 (-2.2, 2.0)	0.6 (-1.5, 2.7)	1.3 (-1.1, 3.7)	1.5 (-0.8, 3.8)	1.9 (-1.1, 4.9)
1958 NCDS	42	4,139	- #	1.0 (0.0, 1.9)	0.8 (-0.1, 1.7)	1.8 (0.8, 2.9)	1.4 (0.5, 2.4)	2.3 (1.1, 3.4)
	44	3,421	- #	0.9 (-0.2, 2.0)	0.6 (-0.5, 1.7)	1.9 (0.7, 3.2)	1.5 (0.4, 2.6)	2.1 (0.7, 3.5)
	50	2,958	-	0.6 (-0.5, 1.7)	0.7 (-0.4, 1.8)	1.8 (0.5, 3.1)	1.3 (0.2, 2.5)	2.7 (1.2, 4.2)
1970 BCS	42	3,386	- #	1.5 (0.7, 2.4)	2.1 (1.2, 3.0)	2.8 (1.7, 3.9)	2.6 (1.7, 3.6)	3.9 (2.3, 5.4)

Table 2. Own social class (42/43 y) and BMI across mid-adulthood in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies

#evidence for deviation from linearity—*p* < 0.05 for Wald tests; BMI differences estimated using linear regression models; occupational-based social classes were assigned as follows: I (professional), II (managerial and technical), III NM (skilled nonmanual), III M (skilled manual), IV (partly-skilled) V (unskilled)

doi:10.1371/journal.pmed.1002214.t002

measured at 42/43 y. To aid comparability of results across cohorts, the Registrar General's Social Class was used to classify social class—from I (professional), II (managerial and technical), IIIN (skilled nonmanual), IIIM (skilled manual), IV (partly-skilled), and V (unskilled) occupations. The 1990 classification schema was used for childhood and adult SEP in all cohorts, and (due to a lack of conversion schema) the 1970 version was used for childhood SEP in the 1946 NSHD. Those in the armed forces were not assigned a social class, nor those not employed.

Analytical strategy

DB, RH, WJ, LL, and DK determined which analyses to perform and include in the paper in August 2016. Following request from peer review, we conducted additional analyses in October 2016 to examine the extent to which associations between childhood SEP and BMI were explained by adult SEP, and to examine in greater detail the characteristics of those with missing data.

SEP differences in BMI. We derived mean BMI (and standard error [SE]) at each age for each childhood and adult SEP group, separately for each cohort and gender. To assess absolute inequalities in BMI at each age for childhood and adult SEP, we applied linear regression models to estimate differences in mean BMI (kg/m²) between each SEP class and the referent group (class I). We additionally checked if results differed when using regression models with log-transformed BMI to estimate relative differences (i.e., percentage) in BMI. To limit the potential for reverse causality, analyses using adult SEP were limited to contemporaneous and future BMI (\geq 42 y). Given expected gender differences in association (with larger inequalities in women),[7, 8] all analyses were conducted separately in each gender, and gender differences were formally tested by including an interaction term (gender*SEP). Deviation from linearity

in the association between SEP and BMI was examined using Wald tests to determine whether coefficients for SEP (modelled as a categorical term) were equal to zero, in a model which also contained SEP modelled as a continuous term. To examine if the size of inequalities in BMI differed by cohort, regression models were also fitted on BMI at 42/43 y for all three cohorts combined, where differences in age of BMI measurements were smallest so that inequalities could be compared with respect to both childhood and adult SEP. A dummy term for each cohort was included in the model, and SEP*cohort interaction was tested. In all cohort-combined models, weights to account for the stratified nature of the 1946 NSHD cohort were applied, and all participants from 1958 NCDS and 1970 BCS were given the same weighting value of one. All analyses were conducted using statistical software STATA 14 (StataCorp, 2009).

SEP differences in risk of overweight or obesity. The primary analyses described above used BMI as a continuous outcome, since preservation of the continuous nature of the outcome preserves statistical power and may enable a greater understanding of the nature of adiposity inequalities than analyses using binary outcomes. However, to aid public health interpretation, all analyses were repeated using a binary outcome indicating normal (BMI < 25) or overweight/obese (BMI \geq 25) as an outcome. Overweight and obese were grouped together given the low obesity prevalence at younger ages, and participants classified as thin were excluded from analyses (2% of observations).[6] Inequalities were estimated using linear probability models to derive differences in prevalence by SEP group, and using log-binomial generalized linear models to estimate the relative risks of overweight/obesity in each SEP group.

Do SEP differences in BMI differ by age?. Trajectories of BMI were modelled using multilevel models—BMI measurements (level 1) were nested within individuals (level 2). We adopted a quadratic function for age to summarize the longitudinal changes of BMI. We specified a random intercept and random slope (linear term for age). SEP (as a categorical variable) was added to the models to examine its associations with BMI across adulthood. Differences in rate of BMI change between SEP groups were examined by including age*SEP interaction terms (age^{2*}SEP interactions were not found and therefore not included in models; p > 0.05). Finally, a cohort combining all three cohorts was fitted, and an age*SEP*cohort interaction term was included to test whether the change in association by age differed by cohort. This model also contained cohort main effects, and all two-way interactions (age*SEP, age*cohort, SEP*cohort). Only fixed effects models were fitted when analysing adult SEP in 1970 BCS, since only one age point of BMI was available. Finally, additional models were conducted to examine whether associations between childhood SEP and BMI were explained by adult SEP. In these models, participants with valid data for SEP and valid BMI data on at least one age were included in analyses.

Additional and sensitivity analyses. To examine the extent to which self-reported BMI data could bias SEP and BMI associations, we calculated differences in BMI in the 1958 NCDS at 42 y (self-reported) and 44 y (objectively measured)—the closest proximity of BMI self-reporting and objective measures in our data. We then examined relations between SEP and this difference measure: larger scores would indicate misreporting and/or excessive weight change. We were unable to adjust for an indicator of self-reporting or objective BMI measurement method in our models due to collinearity between measurement method and age/cohort. Finally, to inform the extent to which differences in missing data between cohorts could affect results, we compared the extent of missing SEP and BMI data by cohort and examined the characteristics of those with missing data.

Results

BMI was typically higher at older ages within each cohort and in cohorts born more recently. BMI was also generally higher amongst those of lower rather than higher SEP (S1 and S2Tables). Obesity or overweight prevalence followed the same patterns (S3 and S4 Tables).

Childhood SEP

In each cohort, lower childhood SEP was associated with higher mean BMI at all ages (Table 1). Associations tended to be stronger among women than men: evidence for gender interaction with childhood SEP (P<0.05) was found in the 1946 NSHD (at 60–64 y), the 1958 NCDS (33, 42, 44, and 50 y), and the 1970 BCS (30 and 42 y). Associations were also found to be nonlinear in many cases, with kg/m² differences in BMI between classes I and IV (partly skilled) being larger than those between I and V (unskilled).

The sizes of SEP differences in BMI were of considerable magnitude. For example, comparing mean BMI at 42/43 y amongst women in the lowest compared with highest childhood SEP, there was a 1.7 kg/m² (95% CI: 0.2, 3.2) difference in the 1946 NSHD, 1.5 kg/m² (0.6, 2.5) difference in the 1958 NCDS, and 2.7 kg/m² (1.6, 3.9) difference in the 1970 BCS. However, there was little evidence that the size of BMI inequalities differed systematically by cohort (*p*value for cohort*childhood SEP term = 0.33 in men and 0.20 in women; findings were similar when analyzing BMI differences in relative instead of absolute scales). Analyses using overweight or obese as a binary outcome yielded similar results (S5 Table). For example, comparing the difference in prevalence at 42/43 y amongst women in the lowest compared with highest childhood SEP, there was a 22.5% (5.9%, 39.2%) difference in the 1946 NSHD, 14.2% (5.1%, 23.3%) in the 1958 NCDS, and 15.7% (5.4%, 25.9%) in the 1970 BCS.

Inequalities in BMI within each cohort were typically larger at older ages, as indicated by positive age*SEP interaction terms for the lower SEP groups (Figs 1 and 2, multilevel model estimates shown in S6 Table). For example, among women in the 1946 NSHD, the estimated BMI difference between class I and V of 1.45 kg/m² (0.61, 2.29) at 26 years had increased to 3.30 kg/m² (1.38, 5.21) at 60–64 y. The size of these age-related increases in inequality did not appear to differ systematically by cohort except in men, where age-related increases appeared to be stronger in the 1958 NCDS and 1970 BCS than the 1946 NSHD (p = 0.001 age*SEP*cohort interaction; p = 0.01 in women). These associations occurred alongside both age-related increases in mean BMI and secular increases in mean BMI in later born cohorts—mean BMI in the highest SEP group in the BCS 1970 was comparable to that of the lowest SEP group in the 1946 NSHD. Finally, associations between childhood SEP and BMI were typically only partly attenuated after adjustment for adult SEP (S7 Table).

Adult SEP

Lower adult SEP was typically associated with higher BMI (Table 2). As with childhood SEP differences, these associations were typically stronger among women than men—evidence for gender interaction with adult SEP (p < 0.05) was found in the 1946 NSHD (at 43, 53, and 60–64 y), the 1958 NCDS (42 and 44 y), and the 1970 BCS (42 y)—and for nonlinearity in the shape of associations.

Among women, the size of BMI inequalities was progressively larger in each subsequent cohort. For example, the mean differences in BMI at 42/43 y amongst women in the lowest compared with highest social class was 2.0 kg/m² (95% CI: -0.1, 4.0) in the 1946 NSHD, 2.3 (1.1, 3.4) in the 1958 NCDS, and 3.9 (2.3, 5.4) in the 1970 BCS—p(for cohort*SEP term) = 0.01. These cohort differences were driven by slight decreases in mean BMI in the highest social class and increases in BMI amongst those in the lowest social classes (S2 Table). There



Fig 1. Male BMI across adulthood in relation to father's social class (10/11 y) in the 1946, 1958, and 1970 British birth cohort studies. Note: lines show estimated BMI along with 95% confidence intervals at each age, estimated using multilevel general linear regression models.

doi:10.1371/journal.pmed.1002214.g001



Fig 2. Female BMI across adulthood in relation to father's social class (10/11 y) in the 1946, 1958, and 1970 British birth cohort studies. Note: lines show estimated BMI along with 95% confidence intervals at each age, estimated using multilevel general linear regression models.

doi:10.1371/journal.pmed.1002214.g002



Fig 3. Male BMI across adulthood in relation to own social class (42/43 y) in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies. Note: lines show estimated BMI along with 95% confidence intervals at each age, estimated using multilevel general linear regression models (age terms not included in the 1970 BCS due to only 1 age of measurement).

doi:10.1371/journal.pmed.1002214.g003

was no such evidence of cohort differences in men (p = 0.7), and findings for either gender did not differ when using relative instead of absolute measures of inequality. For example, among women, percentage differences in BMI at 42/43 in the lowest compared with highest social class were 6.4% (-1.3, 14.1) in the 1946 NSHD, 8.1% (4.0, 12.3) in the 1958 NCDS, and 14.0% (8.5, 19.6) in 1970 BCS. Mean (standard deviation [SD]) BMI in the highest and lowest social classes were as follows: 24.9 (0.8) versus 26.8 (0.7) in the 1946 NSHD, 24.2 (0.4) versus 26.5 (0.4) in the 1958 NCDS, and 24.2 (0.3) versus 28.1 (0.8) in the 1970 BCS (see S2 Table). Analyses using overweight or obese as binary outcomes yielded similar results (S8 Table). For example, when comparing the percentage difference in prevalence at 42/43 y amongst women in the lowest compared with highest adult SEP, there was a 5.2% (-26.2, 36.5) difference in the 1946 NSHD, 20.2% (8.6, 31.9) in the 1958 NCDS, and 26.9% (12.5, 41.2) in the 1970 BCS.

Among men in the 1958 NCDS, adult SEP and BMI associations tended to become larger at older ages—the estimated difference in BMI between class IV and I was 0.98 kg/m² (0.43, 1.54) at 43 y and 1.57 kg/m² (0.91, 2.23) at 50 y (Fig 3 and Fig 4 and S9 Table). These differences were not found among men in the 1946 NSHD (p < 0.001 age*SEP*cohort interaction). Among women, there was no evidence that associations systematically changed with age in either cohort (p = 0.97 age*SEP*cohort interaction).

Additional and sensitivity analyses

We did not find evidence that differences in BMI at 42 (self-reported) and 44 y (objectively measured) differed according to childhood (p = 0.9 in men, p = 0.2 in women) or adult SEP (p = 0.3 in men, p = 0.3 in women) in the 1958 NCDS (S10 Table). More recently born cohorts had greater missing SEP and BMI data; similarly in each cohort, those with lower childhood



Fig 4. Female BMI across adulthood in relation to own social class (42/43 y) in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies. Note: lines show estimated BMI along with 95% confidence intervals at each age, estimated using multilevel general linear regression models (age terms not included in the 1970 BCS due to only one age of measurement).

doi:10.1371/journal.pmed.1002214.g004

SEP or higher preceding BMI were more likely to have missing adult SEP and BMI data (<u>S1</u> Text).

Discussion

Main findings

Using longitudinal data from three British birth cohorts who experienced the obesity epidemic at increasingly younger ages in adulthood, we identified large and persisting socioeconomic inequalities in BMI from age 20 up to ages 60–64. Inequalities according to childhood SEP were evident in both men and women, were typically larger at older ages, and were similar in magnitude at any given age in all three cohorts. Inequalities according to adult SEP were more evident in women than men, yet did not consistently differ by age. Among women, inequalities were found to be larger in both absolute and relative terms in the 1970 BCS compared with earlier-born cohorts in midlife BMI (i.e., BMI at age 42/43 y when measured in 2012 compared with 2000 or 1989), while in men they remained constant.

Comparison with previous studies

Our findings are broadly consistent with evidence of persisting inequalities in obesity or BMI derived from separate studies utilizing repeated cross-sectional data and adult SEP indicators. For example, in English adults aged 18–64, BMI inequalities according to adult manual or nonmanual social class did not change (in absolute terms) from 1993/4 to 2002/3,[13] yet obesity inequalities increased in absolute but not relative terms according to area-based socioeconomic circumstances among older (age \geq 55) but not younger (16–54) adults from 1994 to

2008.[14] In Scotland, obesity inequalities (in relative terms) according to education or income in adults were estimated to have decreased from 1995 to 2011 (age 16–65).[15] Studies using birth cohort data have also reported inequalities in BMI yet are either limited to investigation of single birth cohorts,[22–25, 27] or use multiple birth cohorts in a more limited manner (e.g., among regional samples with limited overlap between ages/y born,[47] or among two national birth cohorts using BMI at one age and a binary SEP indicator[26]). Our findings add to the international evidence base, also derived from repeated cross-sectional data, suggesting persisting inequalities in high income countries and/or emerging inequalities in low–middle income countries.[17–21]

Our study adds to previous knowledge by providing a more detailed understanding of how inequalities in BMI have changed across a longer timeframe. Despite large societal changes in occupations and a reduction in absolute disadvantage in the later 20th century, we found that social class inequalities have persisted across three generations. Two additional findings also suggest a greater potential cause for concern than previously thought: for childhood SEP, associations typically widened with increasing age in all three cohorts and, for adult SEP, associations appear to have become larger in more recently born generations of women.

Explanation of findings

Inequalities in BMI are likely to be explained by inequalities in the determinants of weight gain both before and during adulthood. In support of this, systematic reviews and large population-based studies have found evidence for inequalities in both indicators of physical activity [48, 49] and diet.[50–52] The persistence of inequalities in BMI across cohorts suggests that inequalities in the determinants of BMI have not substantially changed,[53] despite policies designed to reduce them.[54, 55] The net influence of childhood SEP on the determinants of BMI, operating both in childhood and adulthood, may therefore have not differed in each cohort. If this were the case, our finding of increasing size of inequalities in BMI found at older compared with younger ages (according to childhood SEP) may be due to the accumulation of periods of weight gain which would be expected to track across life.[56] Part of the childhood, although we found limited evidence for a prominent mediating role of adult social class (as in other studies[22–26]). However, such pathways may be more appropriately identified in analyses that account for potential intermediary confounding[57] and the multidimensional and time-varying nature of adult SEP.

Although we are not aware of studies that have examined trends in diet and physical activity inequalities in these birth cohorts, there are some co-occurring cultural and economic changes which may have been expected to have increased BMI inequalities across time. For example, the relative cost of a high-quality diet increased in the UK (and other high income countries) from the 1990s–2010,[58] although it is unclear if this was matched by increased inequalities in dietary intake.[14, 59] In addition, studies have suggested that there have been increasing cultural expectations from the 1970s onwards for women to be thin,[60] yet for men to be muscular.[61, 62] Since women of higher SEP are likely to have had greater resources to follow these trends, these changes could explain the increasing inequalities observed in the 1970 BCS compared with earlier born cohorts; mean BMI in the most advantaged group remained similar in all three cohorts. Women also experience additional putative risk factors for weight gain which men do not, such as childbirth and the menopausal transition,[63] yet these factors have not explained BMI inequalities in previous studies.[64] Inequalities in leisure time physical activity widened among older women but not men in England from 1994–2008 (according to an area-based socioeconomic indicator)[14], and occupational-related physical activity may

have declined, especially amongst manual social classes, due to declines in the number of physical occupations.[65] However, it should be noted that understanding the relative contribution of diet and physical activity to BMI inequalities, and how these may change across time, is likely to be challenging due to the difficulties in accurately assessing these behaviours in the population.

In addition, labour market participation has changed across time, with increasing participation from women in the 1970 BCS compared with the 1946 NSHD.[66] Such differences could potentially affect estimates of how inequalities in BMI have changed across time, particularly among women. A conservative interpretation of our results would therefore limit them to providing evidence of how socioeconomic inequality in BMI has changed amongst those whose fathers were employed as children, or those subsequently employed as adults. Despite overall declining population smoking rates, relative inequalities in smoking have increased,[67] a pattern which could affect BMI inequality, since smoking has a modest association with lower weight. However, smoking has been found not to explain associations between SEP and BMI in other cohorts,[68] and smoking was not associated with obesity in the 1946 NSHD,[69] nor in the 1970 BCS.[70] Finally, mortality rates are likely to have been highest in lower SEP groups[71] and those with higher BMI,[72] which might partly explain the lower-thanexpected BMI found in the lowest SEP group among men at older ages.

Strengths and limitations

Strengths of this study include the use of three national birth cohort studies initiated in 1946, 1958, and 1970, with harmonized data for SEP and BMI across life. These data are especially well-suited to examine the emergence of inequalities in BMI across age and generation. Limitations include the use of BMI which, although strongly positively correlated with fat mass[73], does not distinguish fat and lean (muscle) mass. Lower SEP has been associated with lower lean mass in the 1946 NSHD (after adjustment)[23], so the use of BMI could have resulted in underestimating socioeconomic inequalities in fat mass, compared with direct measures of fat mass. Indeed, given evidence that BMI and blood pressure associations are stronger in the 1958 NCDS than the 1946 NSHD,[74] it may be amongst later born cohorts a given BMI value reflects increasingly more fat than lean mass compared with earlier born cohorts.

All included studies experienced attrition that, as expected, [43, 44] was generally more pronounced amongst those of lower SEP and/or with higher BMI. The use of multilevel models enabled those with incomplete information to be included in our analyses (under the assumption of missing at random). However, as in all observational studies, we cannot rule out the possibility that missing data are nonignorable and therefore might bias our findings. Missing SEP and BMI data were also more frequent in later born cohorts, which might bias results regarding cross cohort changes in BMI inequality. Studies have found that greater attrition or nonresponse in longitudinal studies typically leads to a reduction in the magnitude of health inequalities observed; [75] analogous findings were also found in the present study, since childhood SEP and BMI associations were generally weaker in the samples which additionally had valid adult SEP data (S7 Table). This direction of bias may therefore have led to us underestimating increases in BMI inequality observed amongst more recently born cohorts (which were observed for women), and increases in the size of BMI inequality at older ages (which were found for both men and women). In support of the representativeness of the included birth cohorts, obesity estimates have been found to be similar with those from the Health Survey for England.[6] However, such biases should be considered when interpreting these, and other, results on how SEP differences may have changed across time-cross-sectional health surveys have moderate response rates (around 40 to 60%) which have declined in recent

decades.[76] We used harmonized measures of occupational-based social class, yet these cohorts predate the 2000 move to NS-SEC social class used in the UK.[77] The use of social class omitted participants not currently employed; class is also only one dimension of SEP— others (e.g., education, income, and wealth) may yield health-relevant information independent of class, and therefore warrant further study. Finally, we focused on average differences across social groups, whereas within-group differences may have also changed by cohort, and this warrants investigation.[78]

Implications

The persistence of inequalities in BMI throughout adulthood across different generations suggests that new and/or improved strategies are required to reduce them. Indeed, the UK government has, through a number of initiatives, aimed to reduce both obesity and its inequality since 1998, with limited current success. [54, 55] While there is some evidence that interventions targeted at disadvantaged populations can result in reductions in BMI, there is limited high-quality evidence on population-wide interventions, as well as on the relative timing of interventions across life.[79] Given our findings of progressively widening BMI inequalities across adulthood, and the fact that BMI tends to track across life, [56] interventions may be most effective when initiated as early as possible in adulthood. Interventions which require little individual agency^[80] may be especially efficacious at lowering population BMI levels and reducing inequalities. For example, increasing taxation on unhealthy foods while subsidizing others may be effective, despite potentially being financially regressive in the short term.[81, 82] Coordinated analyses of how inequalities in BMI differ across countries may also help to identify strategies which successfully reduce BMI inequalities.[83] Ultimately, targeting inequalities in socioeconomic resources may be more effective than targeting specific mediating factors, since the relative importance of the mediators may change across time and interventions may inadvertently increase inequalities.[84] Reducing inequalities in socioeconomic resources in childhood may be particularly beneficial by affecting both early life determinants of adult BMI and subsequent adult SEP inequalities. Indeed, such early interventions will also be best placed to reduce inequalities in childhood obesity, which are found in cohorts born more recently (e.g., those born in 2000/1);[85] understanding how adiposity inequalities have changed in different generations in childhood requires investigation of childhood growth in terms of weight, height, and height-adjusted weight.

Conclusions

Our findings, based on historic longitudinal data, demonstrate that the overweight and obesity epidemic has disproportionately impacted adults in Britain born in 1946, 1958, and 1970 who were more socioeconomically disadvantaged in childhood or adulthood. They prompt consideration of how inequalities can be reduced amongst these and future cohorts, given the considerable expected adverse health impacts.

Supporting Information

S1 Table. Father's occupational class (10/11 y) and mean BMI across adulthood in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies. (DOC)

S2 Table. Own occupational class (42/43 y) and mean BMI across adulthood in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies. (DOC)

S3 Table. Father's occupational class (10/11 y) and overweight or obesity prevalence across adulthood in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies. (DOC)

S4 Table. Own occupational class (42/43 y) and overweight or obesity prevalence across adulthood in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies. (DOC)

S5 Table. Father's occupational class (10/11 y) and obesity or overweight across adulthood in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies. (DOC)

S6 Table. Father's occupational class (10/11 y) and BMI across adulthood (\geq 20 y) in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies: estimates from separate multilevel models, scaled to show estimated BMI differences at 26 y. (DOC)

S7 Table. Father's occupational class (10/11 y) and BMI across adulthood (\geq 20 y) in the 1946 NSHD, 1958 NCDS, and 1970 BCS British birth cohort studies, adjusted for adult occupational class (42/43 y): estimates from separate multilevel models, scaled to show estimated BMI differences at 26 y. (DOC)

S8 Table. Own occupational class (42/43 y) and obesity or overweight across mid-adulthood in the 1946 NSHD, 1958 NCDS, and 1970 BCS birth cohort studies. (DOC)

S9 Table. Own occupational class (42/43 y) and adult BMI (\geq 42 y) in the 1946 NSHD and 1958 NCDS British birth cohort studies: estimates from separate multilevel models, scaled to show estimated BMI differences at 43 y. (DOC)

S10 Table. Socioeconomic position in relation to calculated change in BMI between a selfreported (at 42 y) and objective measure (at 44 y) in the 1958 NCDS British birth cohort study

(DOC)

S1 PRIMSA Checklist.(DOC)S1 Text. Missing data appendix.

(DOC)

Acknowledgments

We thank Dr. Shaun Scholes for providing helpful comments on an earlier version of this manuscript and Brian Dodgeon for preparing harmonized social class data. We thank participants from all three studies for their invaluable long-standing contribution.

Author Contributions

Conceptualization: DB RH WJ LL DK.

Data curation: DB WJ RH.

Formal analysis: DB.

Funding acquisition: LL DK RH.

Methodology: DB RH WJ LL DK.

Project administration: DB RH LL DK.

Software: DB.

Supervision: RH.

Visualization: DB.

Writing - original draft: DB.

Writing - review & editing: DB RH WJ LL DK.

References

- 1. Withrow D, Alter D. The economic burden of obesity worldwide: a systematic review of the direct costs of obesity. Obes Rev. 2011; 12(2):131–41. doi: 10.1111/j.1467-789X.2009.00712.x PMID: 20122135
- 2. Global BMI Mortality Collaboration. Body-mass index and all-cause mortality: individual-participantdata meta-analysis of 239 prospective studies in four continents. Lancet. 2016; 20(88):776–86.
- Ng M, Fleming T, Robinson M, Thomson B, Graetz N, Margono C, et al. Global, regional, and national prevalence of overweight and obesity in children and adults during 1980–2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet. 2014; 384(9945):766–81. doi: 10.1016/S0140-6736(14)60460-8 PMID: 24880830
- 4. Finucane MM, Stevens GA, Cowan MJ, Danaei G, Lin JK, Paciorek CJ, et al. National, regional, and global trends in body-mass index since 1980: systematic analysis of health examination surveys and epidemiological studies with 960 country-years and 9.1 million participants. Lancet. 2012; 377 (9765):557–67.
- NCD Risk Factor Collaboration. Trends in adult body-mass index in 200 countries from 1975 to 2014: a pooled analysis of 1698 population-based measurement studies with 19.2 million participants. Lancet. 2016; 387(10026):1377–96. doi: 10.1016/S0140-6736(16)30054-X PMID: 27115820
- Johnson W, Li L, Kuh D, Hardy R. How Has the Age-Related Process of Overweight or Obesity Development Changed over Time? Co-ordinated Analyses of Individual Participant Data from Five United Kingdom Birth Cohorts. PLoS Med. 2015; 12(5):e1001828. doi: 10.1371/journal.pmed.1001828 PMID: 25993005
- McLaren L. Socioeconomic status and obesity. Epidemiol Rev. 2007; 29:29–48. doi: 10.1093/epirev/ mxm001 PMID: 17478442
- Senese LC, Almeida ND, Fath AK, Smith BT, Loucks EB. Associations Between Childhood Socioeconomic Position and Adulthood Obesity. Epidemiol Rev. 2009; 31(1):21–51.
- Shrewsbury V, Wardle J. Socioeconomic Status and Adiposity in Childhood: A Systematic Review of Cross-sectional Studies 1990–2005. Obesity. 2008; 16(2):275–84. doi: 10.1038/oby.2007.35 PMID: 18239633
- 10. The Marmot Review Team. Fair Society, Healthy Lives: The strategic review of health inequalities in England, post 2010. 2010
- Stringhini S S. ASsociation of socioeconomic position with health behaviors and mortality. JAMA. 2010; 303(12):1159–66. doi: 10.1001/jama.2010.297 PMID: 20332401
- World Health Organization. Obesity and inequities: Guidance for addressing inequities in overweight and obesity. Available from: www.euro.who.int/__data/assets/pdf_file/0003/247638/obesity-090514. pdf. 2014.
- **13.** Wardle J, Boniface D. Changes in the distributions of body mass index and waist circumference in English adults, 1993/1994 to 2002/2003. Int J Obes. 2008; 32(3):527–32.
- Scholes S, Bajekal M, Love H, Hawkins N, Raine R, O'Flaherty M, et al. Persistent socioeconomic inequalities in cardiovascular risk factors in England over 1994–2008: A time-trend analysis of repeated cross-sectional data. BMC Public Health. 2012; 12(1):129.

- Zhu J, Coombs N, Stamatakis E. Temporal trends in socioeconomic inequalities in obesity prevalence among economically-active working-age adults in Scotland between 1995 and 2011: a populationbased repeated cross-sectional study. BMJ Open. 2015;5(6).
- Frederick CB, Snellman K, Putnam RD. Increasing socioeconomic disparities in adolescent obesity. Proc Natl Acad Sci USA. 2014; 11(4):1338–42.
- Harper S, Lynch J. Trends in socioeconomic inequalities in adult health behaviors among US states, 1990–2004. Public Health Rep. 2007; 122(2):177. PMID: 17357360
- Hajizadeh M, Karen Campbell M, Sarma S. Socioeconomic inequalities in adult obesity risk in Canada: trends and decomposition analyses. The European Journal of Health Economics. 2014; 15(2):203–21. doi: 10.1007/s10198-013-0469-0 PMID: 23543117
- Chang VW, Lauderdale DS. Income disparities in body mass index and obesity in the United States, 1971–2002. Arch Intern Med. 2005; 165(18):2122–8. doi: 10.1001/archinte.165.18.2122 PMID: 16217002
- Norberg M, Lindvall K, Stenlund H, Lindahl B. The obesity epidemic slows among the middle-aged population in Sweden while the socioeconomic gap widens. Global health action. 2010; 3.
- Jones-Smith JC, Gordon-Larsen P, Siddiqi A, Popkin BM. Cross-national comparisons of time trends in overweight inequality by socioeconomic status among women using repeated cross-sectional surveys from 37 developing countries, 1989–2007. Am J Epidemiol. 2011; 173(6):667–75. doi: <u>10.1093/aje/ kwg428 PMID: 21300855</u>
- Birnie K, Cooper R, Martin RM, Kuh D, Sayer AA, Alvarado BE, et al. Childhood Socioeconomic Position and Objectively Measured Physical Capability Levels in Adulthood: A Systematic Review and Meta-Analysis. PLoS ONE. 2011; 6(1):e15564. doi: 10.1371/journal.pone.0015564 PMID: 21297868
- Bann D, Cooper R, Wills AK, Adams J, Kuh D. Socioeconomic position across life and body composition in early old age: findings from a British birth cohort study. J Epidemiol Community Health. 2014; 68 (6):516. doi: 10.1136/jech-2013-203373 PMID: 24567442
- Power C, Manor O, Matthews S. Child to adult socioeconomic conditions and obesity in a national cohort. Int J Obes Relat Metab Disord. 2003; 27(9):1081–6. doi: 10.1038/sj.ijo.0802323 PMID: 12917714
- Strand B, Murray ET, Guralnik J, Hardy R, Kuh D. Childhood social class and adult adiposity and bloodpressure trajectories 36–53 years: gender-specific results from a British birth cohort. J Epidemiol Community Health. 2012; 66(6):512–8. doi: 10.1136/jech.2010.115220 PMID: 21098826
- Power C, Graham H, Due P, Hallqvist J, Joung I, Kuh D, et al. The contribution of childhood and adult socioeconomic position to adult obesity and smoking behaviour: an international comparison. Int J Epidemiol. 2005; 34(2):335–44. doi: 10.1093/ije/dyh394 PMID: 15659473
- Langenberg C, Hardy R, Kuh D, Brunner E, Wadsworth M. Central and total obesity in middle aged men and women in relation to lifetime socioeconomic status: evidence from a national birth cohort. J Epidemiol Community Health. 2003; 57(10):816–22. doi: 10.1136/jech.57.10.816 PMID: 14573589
- Baum CL, Ford WF. The wage effects of obesity: a longitudinal study. Health Econ. 2004; 13(9):885– 99. doi: 10.1002/hec.881 PMID: 15362180
- 29. Tyrrell J, Jones SE, Beaumont R, Astley CM, Lovell R, Yaghootkar H, et al. Height, body mass index, and socioeconomic status: mendelian randomisation study in UK Biobank. BMJ. 2016;352.
- Lundborg P, Nystedt P, Rooth D-O. Body Size, Skills, and Income: Evidence From 150,000 Teenage Siblings. Demography. 2014; 51(5):1573–96. doi: 10.1007/s13524-014-0325-6 PMID: 25199549
- Kinge JM. Body mass index and employment status: A new look. Econ Hum Biol. 2016; 22:117–25. doi: 10.1016/j.ehb.2016.03.008 PMID: 27054482
- King NB, Harper S, Young ME. Use of relative and absolute effect measures in reporting health inequalities: structured review. BMJ. 2012;345.
- Mackenbach JP, Martikainen P, Menvielle G, de Gelder R. The arithmetic of reducing relative and absolute inequalities in health: a theoretical analysis illustrated with European mortality data. J Epidemiol Community Health. 2016; 70(7):730–6. doi: 10.1136/jech-2015-207018 PMID: 26945094
- Mackenbach JP, Kulhánová I, Bopp M, Deboosere P, Eikemo TA, Hoffmann R, et al. Variations in the relation between education and cause-specific mortality in 19 European populations: A test of the fundamental causes theory of social inequalities in health. Soc Sci Med. 2014; 127:51. doi: 10.1016/j. socscimed.2014.05.021 PMID: 24932917
- Phelan JC, Link BG, Tehranifar P. Social Conditions as Fundamental Causes of Health Inequalities: Theory, Evidence, and Policy Implications. J Health Soc Behav. 2010; 51(1 suppl):S28–S40. doi: 10. 1177/0022146510383498 PMID: 20943581

- 36. Kuh D, M P, J A, J D, U E, P F, et al. Updating the cohort profile for the MRC National Survey of Health and Development: a new clinic-based data collection for ageing research. Int J Epidemiol. 2011; 40(1): e1–e9. doi: 10.1093/ije/dyq231 PMID: 21345808
- Wadsworth M, Kuh D, Richards M, Hardy R. Cohort profile: The 1946 National Birth Cohort (MRC National Survey of Health and Development). Int J Epidemiol. 2006; 35(1):49–54. doi: 10.1093/ije/ dyi201 PMID: 16204333
- Power C, Elliott J. Cohort profile: 1958 British birth cohort (National Child Development Study). Int J Epidemiol. 2006; 35(1):34–41. doi: 10.1093/ije/dyi183 PMID: 16155052
- Elliott J, Shepherd P. Cohort profile: 1970 British birth cohort (BCS70). Int J Epidemiol. 2006; 35 (4):836–43. doi: 10.1093/ije/dyl174 PMID: 16931528
- Wadsworth M. The origins and innovatory nature of the 1946 British national birth cohort study. Longitudinal and Life Course Studies. 2010; 1(2):121–36.
- **41.** Wadsworth ME, Bynner J. A Companion to Life Course Studies: the social and historical context of the British birth cohort studies: Taylor & Francis; 2011.
- 42. Wadsworth M. The imprint of time: Childhood, history and adult life. Oxford, UK: Clarendon Press; 1991 1991.
- 43. Stafford M, Black S, Shah I, Hardy R, Pierce M, Richards M, et al. Using a birth cohort to study ageing: representativeness and response rates in the National Survey of Health and Development. European Journal of Ageing. 2013; 10:145–57. doi: 10.1007/s10433-013-0258-8 PMID: 23637643
- 44. Mostafa T, Wiggins RD. The impact of attrition and non-response in birth cohort studies: a need to incorporate missingness strategies. Longitudinal and Life Course Studies. 2015; 6(2):131–46.
- Atherton K, Fuller E, Shepherd P, Strachan D, Power C. Loss and representativeness in a biomedical survey at age 45 years: 1958 British birth cohort. J Epidemiol Community Health. 2008; 62(3):216–23. doi: 10.1136/jech.2006.058966 PMID: 18272736
- 46. Wadsworth ME, Mann SL, Rodgers B, Kuh DJ, Hilder WS, Yusuf EJ. Loss and representativeness in a 43 year follow up of a national birth cohort. J Epidemiol Community Health. 1992; 46(3):300–4. PMID: 1645091
- 47. Shaw RJ, Green MJ, Popham F, Benzeval M. Differences in adiposity trajectories by birth cohort and childhood social class: evidence from cohorts born in the 1930s, 1950s and 1970s in the west of Scotland. J Epidemiol Community Health. 2014.
- Elhakeem A, Cooper R, Bann D, Hardy R. Childhood socioeconomic position and adult leisure-time physical activity: a systematic review. International Journal of Behavioral Nutrition and Physical Activity. 2015; 12(1):92.
- Gidlow C, Johnston LH, Crone D, Ellis N, James D. A systematic review of the relationship between socio-economic position and physical activity. Health Educ J. 2006; 65(4):338–67.
- Pechey R, Jebb SA, Kelly MP, Almiron-Roig E, Conde S, Nakamura R, et al. Socioeconomic differences in purchases of more vs. less healthy foods and beverages: analysis of over 25,000 British households in 2010. Soc Sci Med. 2013; 92:22–6. doi: 10.1016/j.socscimed.2013.05.012 PMID: 23849275
- Giskes K, Avendano M, rug J, unst AE. A systematic review of studies on socioeconomic inequalities in dietary intakes associated with weight gain and overweight/obesity conducted among European adults. Obes Rev. 2010; 11(6):413–29. doi: 10.1111/j.1467-789X.2009.00658.x PMID: 19889178
- Maguire ER, Monsivais P. Socio-economic dietary inequalities in UK adults: an updated picture of key food groups and nutrients from national surveillance data. Br J Nutr. 2015; 113(01):181–9.
- Bartley M, Fitzpatrick R, Firth D, Marmot M. Social distribution of cardiovascular disease risk factors: change among men in England 1984–1993. J Epidemiol Community Health. 2000; 54(11):806–14. doi: 10.1136/jech.54.11.806 PMID: 11027193
- Mackenbach JP. Can we reduce health inequalities? An analysis of the English strategy (1997–2010). J Epidemiol Community Health. 2011; 65(7):568–75. doi: 10.1136/jech.2010.128280 PMID: 21459930
- Butland B, Jebb S, Kopelman P, McPherson K, Mardell J, Parry V. Foresight: Tackling Obesities: Future Choices—Project report. London, UK: 2007.
- 56. Singh AS, Mulder C, Twisk JWR, van Mechelen W, Chinapaw MJM. Tracking of childhood overweight into adulthood: a systematic review of the literature. Obes Rev. 2008; 9(5):474–88. doi: 10.1111/j.1467-789X.2008.00475.x PMID: 18331423
- Nandi A, Glymour MM, Kawachi I, VanderWeele TJ. Using marginal structural models to estimate the direct effect of adverse childhood social conditions on onset of heart disease, diabetes, and stroke. Epidemiology (Cambridge, Mass). 2012; 23(2):223.

- Wiggins S, Keats S, Han E, Shimokawa S, Alberto J, Hernandez V, et al. The Rising Cost of a Healthy Diet: Changing Relative Prices of Foods in High-Income and Emerging Economies. Overseas Development Institute Report May. 2015; 67.
- Barton KL, Wrieden WL, Sherriff A, Armstrong J, Anderson AS. Trends in socio-economic inequalities in the Scottish diet: 2001–2009. Public Health Nutr. 2015; 18(16):2970–80. doi: <u>10.1017/</u> S1368980015000361 PMID: 25771827
- 60. Swami V. Cultural influences on body size ideals. Eur Psychol. 2015; 20:44–51.
- Pope HG, Olivardia R, Gruber A, Borowiecki J. Evolving ideals of male body image as seen through action toys. Int J Eat Disord. 1999; 26(1):65–72. PMID: 10349585
- Leit RA, Pope HG, Gray JJ. Cultural expectations of muscularity in men: The evolution of Playgirl centerfolds. Int J Eat Disord. 2001; 29(1):90–3. PMID: 11135340
- Wells JCK, Marphatia AA, Cole TJ, McCoy D. Associations of economic and gender inequality with global obesity prevalence: Understanding the female excess. Soc Sci Med. 2012; 75(3):482–90. doi: 10.1016/j.socscimed.2012.03.029 PMID: 22580078
- Hardy R, Wadsworth M, Kuh D. The influence of childhood weight and socioeconomic status on change in adult body mass index in a British national birth cohort. Int J Obes. 2000; 24(6):725–34.
- 65. Buckley JP, Hedge A, Yates T, Copeland RJ, Loosemore M, Hamer M, et al. The sedentary office: a growing case for change towards better health and productivity. Expert statement commissioned by Public Health England and the Active Working Community Interest Company. Br J Sports Med. 2015.
- 66. Ferri E, Bynner J, Wadsworth M. Changing Britain, changing lives: Institute of Education Press; 2003.
- **67.** Davy M. Socio-economic inequalities in smoking: an examination of generational trends in Great Britain. Health Statistics Quarterly. 2007;(34):26. PMID: 17580645
- Heraclides A, Witte D, Brunner E. The association between father's social class and adult obesity is not explained by educational attainment and an unhealthy lifestyle in adulthood. Eur J Epidemiol. 2008; 23 (8):573–9. doi: 10.1007/s10654-008-9245-3 PMID: 18409006
- Braddon FFM, Rodgers B, Wadsworth MEJ, Davies JMC. Onset of Obesity in A 36-Year Birth Cohort Study. Br Med J. 1986; 293(6542):299–303.
- Crawley H, While D. The diet and body weight of British teenage smokers at 16–17 years. Eur J Clin Nutr. 1995; 49(12):904–14. PMID: 8925792
- Kuh D, Hardy R, Langenberg C, Richards M, Wadsworth MEJ. Mortality in adults aged 26–54 years related to socioeconomic conditions in childhood and adulthood: post war birth cohort study. BMJ. 2002; 325(7372):1076–80. PMID: 12424168
- 72. Strand BH, Kuh D, Shah I, Guralnik J, Hardy R. Childhood, adolescent and early adult body mass index in relation to adult mortality: results from the British 1946 birth cohort. J Epidemiol Community Health. 2010; 66(3):225–32. doi: 10.1136/jech.2010.110155 PMID: 20889586
- 73. Flegal KM, Shepherd JA, Looker AC, Graubard BI, Borrud LG, Ogden CL, et al. Comparisons of percentage body fat, body mass index, waist circumference, and waist-stature ratio in adults. The American Journal of Clinical Nutrition. 2009; 89(2):500–8. doi: 10.3945/ajcn.2008.26847 PMID: 19116329
- Li L, Hardy R, Kuh D, Power C. Life-course body mass index trajectories and blood pressure in mid life in two British birth cohorts: stronger associations in the later-born generation. Int J Epidemiol. 2015; 44 (3):1018–26. doi: 10.1093/ije/dyv106 PMID: 26078389
- 75. Howe LD, Tilling K, Galobardes B, Lawlor DA. Loss to follow-up in cohort studies: bias in estimates of socioeconomic inequalities. Epidemiology. 2013; 24(1):1–9. doi: 10.1097/EDE.0b013e31827623b1 PMID: 23211345
- 76. Mindell JS, Giampaoli S, Goesswald A, Kamtsiuris P, Mann C, Männistö S, et al. Sample selection, recruitment and participation rates in health examination surveys in Europe–experience from seven national surveys. BMC Med Res Methodol. 2015; 15(1):1.
- 77. Rose D, Pevalin D, O'Reilly K. The National Statistics Socio-economic Classification: Origins, Development and Use. Basingstoke, UK: 2005 2005. Report No.
- Krishna A, Razak F, Lebel A, Davey Smith G, Subramanian S. Trends in group inequalities and interindividual inequalities in BMI in the United States, 1993–2012. The American Journal of Clinical Nutrition. 2015; 101(3):598–605. doi: 10.3945/ajcn.114.100073 PMID: 25733645
- **79.** Bambra C, Hillier F, Cairns J, Kasim A, Moore H, Summerbell C. How effective are interventions at reducing socioeconomic inequalities in obesity among children and adults? two systematic reviews. Public health research. 2015; 3(1).
- Adams J, Mytton O, White M, Monsivais P. Why Are Some Population Interventions for Diet and Obesity More Equitable and Effective Than Others? The Role of Individual Agency. PLoS Med. 2016; 13(4): e1001990. doi: 10.1371/journal.pmed.1001990 PMID: 27046234

- 81. Mytton OT, Clarke D, Rayner M. Taxing unhealthy food and drinks to improve health. BMJ. 2012;344.
- 82. Powell LM, Chriqui JF, Khan T, Wada R, Chaloupka FJ. Assessing the potential effectiveness of food and beverage taxes and subsidies for improving public health: a systematic review of prices, demand and body weight outcomes. Obes Rev. 2013; 14(2):110–28. doi: 10.1111/obr.12002 PMID: 23174017
- Mackenbach JP, Kulhánová I, Bopp M, Borrell C, Deboosere P, Kovács K, et al. Inequalities in Alcohol-Related Mortality in 17 European Countries: A Retrospective Analysis of Mortality Registers. PLoS Med. 2015; 12(12):e1001909. doi: 10.1371/journal.pmed.1001909 PMID: 26625134
- Lorenc T, Petticrew M, Welch V, Tugwell P. What types of interventions generate inequalities? Evidence from systematic reviews. J Epidemiol Community Health. 2013; 67(2):190–3. doi: 10.1136/jech-2012-201257 PMID: 22875078
- Shackleton N, Hale D, Viner RM. Trends and socioeconomic disparities in preadolescent's health in the UK: evidence from two birth cohorts 32 years apart. J Epidemiol Community Health. 2015; 70(2):140–6. doi: 10.1136/jech-2015-205603 PMID: 26359506